Author's response to reviews

**Title:** Effect of changed organisation of nutritional care of Danish medical in-patients

**Authors:**

Karin O Lassen (k.lassen@dadlnet.dk)
Edvin Grinderslev (edvin@grinderslev.org)
Ruth Nyholm (ruthiohnn@hotmail.dk)

**Version:** 5  **Date:** 14 July 2008

**Author's response to reviews:** see over
Dear Editorial Team,

Thank you for the review of our manuscript “Effect of changed organisation of nutritional care of Danish medical inpatients”.

The revised manuscript conforms to all the comments mentioned of the Editorial Team, and the response to the comments is given below:

- It is described in the manuscript that the qualitative methods used to investigate the effect of the intervention experienced by the patients and the staff fulfilled the Declaration of Helsinki II.
- It is described in the manuscript that the patients and staff received oral information about the investigation underlining the voluntary and anonymous nature of their participation.
- The quantitative methods used in this study are identical to the methods used in previous studies (see ref 6, 26,36). It is described, that these quantitative methods, use of factual and describing data fell outside the framework of the Local Scientific Ethics Committee as concluded by the Local Scientific Ethics Committee of the County of Aarhus, Denmark.
- Referee 2 has no concern, and express that he is satisfied with the manuscript.

We hope that the revised manuscript satisfy the comments you have addressed. We all agree about the changes in the reviewed manuscript, and we are looking forward to your reply.

Yours sincerely

Karin O. Lassen
Ruth Nyholm
Edvin Grinderslev